

Author's response to reviews

Title: Prospective study of urinary tract infection surveillance after kidney transplantation

Authors:

Roberto Ribera-Sanchez (cerdea321@yahoo.com.mx)
Dolores Delgado-Ochoa (lolgera@hotmail.com)
Ramon Espinoza-Hernández (espinohdzt2003@yahoo.com.mx)
Rocio Flores-Paz (rflorespaz@yahoo.com.mx)
Elvia García-Jiménez (elviagen@yahoo.com.mx)
Andres Bazan-Burges (bazandr@hotmail.com)
Myriam Arriaga-Alba (arriaga_alba@yahoo.com)

Version: 9 **Date:** 18 June 2010

Author's response to reviews:

June 18, 2010.

Rachel Neilan, MSc

Scientific Editor

BMC-series Journals

BioMed Central

Floor 6, 236 Gray's Inn Road

London, WC1X 8HL

MS: 1833292702285714

Prospective study of urinary tract infection surveillance after kidney transplantation

Roberto Ribera-Sanchez, Dolores Delgado-Ochoa, Ramon Espinoza-Hernández, Rocio Flores-Paz, Elvia García-Jiménez, Andres Bazan-Burges and Myriam Arriaga-Alba

Dear Rachel Neilan, MSc.

We thank your letter and your comments to improve our work Prospective study of urinary tract infection surveillance after kidney transplantation. With the present, we are including onto the Methods section, a new section remarking the ethical considerations and the name of the comities, responsible of evaluating the research studies which are performed onto our institution, Hospital Juárez of

México.

Yours faithfully,

PhD. Myriam Arriaga-Alba.